Author's response to reviews

Title: Patient- and provider-level risk factors associated with default from tuberculosis treatment, South Africa, 2002: a case-control study

Authors:

Alyssa Finlay (avf0@cdc.gov)  
Joey Lancaster (joey.lancaster@mrc.ac.za)  
Timothy H Holtz (tkh3@cdc.gov)  
Karin Weyer (weyerk@who.int)  
Abe Miranda (aci5@cdc.gov)  
Martie van der Walt (martie.van.der.walt@mrc.ac.za)

Version: 14 Date: 18 January 2012

Author's response to reviews:

Dear editors,

I have fixed some spacing problems in Tables 2, 3, and 4, and have taken out one sentence from our "Footnotes" section that ended up in your Competing Interests Section. That is all.

Thank you.